

GOPEN ACCESS

Citation: Koné M, N'Gouan EK, Kaba D, Koffi M, Kouakou L, N'Dri L, et al. (2020) The complex health seeking pathway of a human African trypanosomiasis patient in Côte d'Ivoire underlines the need of setting up passive surveillance systems. PLoS Negl Trop Dis 14(9): e0008588. https://doi.org/10.1371/journal.pntd.0008588

Editor: Enock Matovu, Makerere University, UGANDA

Received: March 23, 2020

Accepted: July 13, 2020

Published: September 14, 2020

Copyright: © 2020 Koné et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This study was funded by the EDCTP2 programme supported by the European Union (grant number DRIA-2014-306-DiTECT-HAT). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

RESEARCH ARTICLE

The complex health seeking pathway of a human African trypanosomiasis patient in Côte d'Ivoire underlines the need of setting up passive surveillance systems

Minayégninrin Koné^{1,2©}, Emmanuel Kouassi N'Gouan^{3©}, Dramane Kaba¹, Mathurin Koffi[®]², Lingué Kouakou⁴, Louis N'Dri¹, Cyrille Mambo Kouamé¹, Valentin Kouassi Nanan⁵, Gossé Apollinaire Tapé⁶, Bamoro Coulibaly¹, Fabrice Courtin^{1,7}, Bernardin Ahouty[®]², Vincent Djohan¹, Bruno Bucheton⁷, Philippe Solano⁷, Philippe Büscher⁸, Veerle Lejon⁷, Vincent Jamonneau^{1,7}*

1 Unité de Recherche « Trypanosomoses », Institut Pierre Richet, Bouaké, Côte d'Ivoire, 2 Laboratoire de Biodiversité et Gestion des Ecosystèmes Tropicaux, Unité de Recherche en Génétique et Epidémiologie Moléculaire, Université Jean Lorougnon Guédé, UFR Environnement, Daloa, Côte d'Ivoire, 3 Projet de Recherches Cliniques sur la Trypanosomiase (PRCT), Daloa, Côte d'Ivoire, 4 Programme National d'Élimination de la Trypanosomose Humaine Africaine, Abidjan, Côte d'Ivoire, 5 Direction Départementale de la Marahoué, District sanitaire de Sinfra, Ministère de la Santé et de l'Hygiène Publique, Abidjan, Côte d'Ivoire, 6 Direction départementale de la santé de la Marahoué, Centre de Santé Urbain de Bonon, Ministère de la Santé et de l'Hygiène Publique, Abidjan Côte d'Ivoire, 7 Unité Mixte de Recherche IRD-CIRAD 177, INTERTRYP, Institut de Recherche pour le Développement (IRD) Université de Montpellier, Montpellier, France, 8 Department of Biomedical Sciences, Institute of Tropical Medicine, Antwerp, Belgium

These authors contributed equally to this work.
* vincent.jamonneau@ird.fr

Abstract

Background

Significant efforts to control human African trypanosomiasis (HAT) over the two past decades have resulted in drastic decrease of its prevalence in Côte d'Ivoire. In this context, passive surveillance, integrated in the national health system and based on clinical suspicion, was reinforced. We describe here the health-seeking pathway of a girl who was the first HAT patient diagnosed through this strategy in August 2017.

Methods

After definitive diagnosis of this patient, epidemiological investigations were carried out into the clinical evolution and the health and therapeutic itinerary of the patient before diagnosis.

Results

At the time of diagnosis, the patient was positive in both serological and molecular tests and trypanosomes were detected in blood and cerebrospinal fluid. She suffered from important neurological disorders. The first disease symptoms had appeared three years earlier, and the patient had visited several public and private peripheral health care centres and

Competing interests: The authors have declared that no competing interests exist.

hospitals in different cities. The failure to diagnose HAT for such a long time caused significant health deterioration and was an important financial burden for the family.

Conclusion

This description illustrates the complexity of detecting the last HAT cases due to complex diagnosis and the progressive disinterest and unawareness by both health professionals and the population. It confirms the need of implementing passive surveillance in combination with continued sensitization and health staff training.

Author summary

Human African trypanosomiasis (HAT) or sleeping sickness is a parasitic disease caused by *Trypanosoma brucei* that is transmitted by tsetse flies. In 2012, HAT was included in the World Health Organization roadmap for the control of neglected tropical diseases with the objective of elimination as a public health problem by 2020. In Côte d'Ivoire, HAT prevalence has dropped sharply the last decade. A passive HAT surveillance was therefore integrated in the national health system, which allowed to detect a first patient in 2017. This article describes the complex health seeking pathway and suffering before diagnosis of this patient, an 11 years old girl, and illustrates the challenge when health agents and population no longer consider HAT as a threat in an elimination context. Our results show the need to install a solid surveillance system, in combination with continued sensitization and repeated health staff training.

Introduction

Human African trypanosomiasis (HAT) or sleeping sickness is a neglected tropical disease caused by the protozoan parasites *Trypanosoma brucei gambiense* (*Tbg*) and *Tb rhodesiense* (*Tbr*) which are transmitted by tsetse fly (*Glossina spp*) bites. The disease occurs in Sub-Saharan Africa, within the limits of its vector distribution [1]. Thanks to the intensive control efforts deployed since the early 2000s, the number of HAT cases has evolved from 26550 in the year 2000 to 977 in 2018. However, still 57 million people are at risk for infection. As defined in its roadmap, WHO has targeted HAT elimination as a public health problem by the year 2020, and interruption of transmission by 2030 [2].

Côte d'Ivoire is at present the second HAT affected country in West-Africa, with endemic foci in Central-Western forest areas [3,4]. Rural populations living from handwork in cash (cocoa, coffee) and subsistence (banana, rice) agriculture are the most affected. For a long time, the HAT control strategy in Côte d'Ivoire has been based on active case detection by mass screening using the Card Agglutination Test for Trypanosomiasis (CATT) [5], followed by parasitological examination of seropositive individuals [6]. This strategy, implemented by mobile teams who travel from village to village, has allowed to drastically reduce the number of detected HAT cases, with less than 10 yearly cases since 2009 [2,4,7].

However, in a low prevalence context, this strategy is not cost effective. The population does not consider HAT as a threat anymore and refuses to participate to mass screening activities [8,9]. The fact that in 2010 and 2011 not a single case of HAT was found during active screening, while 12 HAT patients were diagnosed in only one national treatment centre [7], demonstrates well the need of implementing a passive HAT surveillance based on clinical

suspicion integrated in the national health system [10,11,12]. This need is also confirmed by the itinerary of a HAT patient that we describe here, illustrating the progressive disinterest and unawareness by both health professionals and the population as HAT is being eliminated.

Materials and methods

Study area

The national health system in Côte d'Ivoire is organized as a classical three level pyramid, with health centres that serve as an entry point at the basis, general hospitals at the secondary level and university hospitals and specialized institutes at the tertiary level. These public health structures are complemented by private clinics and hospitals as well as a network of traditional medicine officially recognized by the Ministry of Health[13]. Until 2017, passive HAT surveil-lance in Côte d'Ivoire relied on two HAT treatment centres: the Projet de Recherches Cliniques sur la Trypanosomiase in Daloa (PRCT) and the Base de Santé Rurale in Bouaflé (closed in 2006) (Fig 1). In August 2017, passive surveillance was extended to the two last endemic foci of the country, Bonon and Sinfra, within the framework of a diagnostic clinical trial (DiTECT-HAT) evaluating the performance of diagnostic tools and algorithm. Ten health



Fig 1. Map showing the localisation of the passive surveillance sites implemented in the Bonon and Sinfra HAT endemic foci in Côte d'Ivoire in 2017.

https://doi.org/10.1371/journal.pntd.0008588.g001

structures were selected at the primary level (Fig 1). Two structures, the Sinfra Hôpital Général (HG) and the Bonon Centre de Santé Urbain (CSU), were mandated to perform clinical and serological screening as well as parasitological confirmation and treatment (centres for diagnosis and treatment). The eight other were health centres performed clinical and serological screening only, and referred seropositive suspects to one of the two centres for confirmation diagnosis.

Diagnostic methods for passive surveillance in Bonon and Sinfra

The diagnostic procedure was initiated by clinical suspicion for HAT. The following symptoms were considered: long-lasting fever resistant to anti-malaria treatment, long-lasting headaches (>14 days), swollen cervical lymph nodes, weight loss, weakness, intense pruritus, amenorrhea, abortion, sterility, coma, psychiatric problems (aggressiveness, apathy, mental confusion, hilarity), sleep disruption (nocturnal insomnia and excessive daytime sleepiness), motor problems (convulsions, abnormal movements, tremor, walking difficulties) and speech difficulties. An individual was a clinical suspect if at least one of these symptoms was present, which initiated serologically testing for presence of specific antibodies. Within the DiTECT-HAT diagnostic trial, three rapid diagnostic tests (RDT) were carried out simultaneously: SD Bioline HAT 1.0 (Standard Diagnostics, Korea, [14]), HAT Sero-K-Set (Coris Bioconcept, Belgium, [15]) and rHAT Sero-Strip (Coris Bioconcept, Belgium, [16]). Clinical suspects positive in at least one RDT, underwent parasitological examination by the mini-anion exchange centrifugation technique (mAECT) performed on whole blood [17, 18] and for those who had swollen cervical lymph nodes, a microscopic examination (x400 magnification) of lymph node fluid. Dried blood spots were also prepared for further testing in immunotrypanolysis [19], enzymelinked immunosorbent assay (ELISA/Tbg [20,21]), loop-mediated isothermal amplification (LAMP [22]), and real-time PCR targeting the 18S rRNA and the Trypanosoma brucei gambiense-specific glycoprotein (TgsGP) genes specific respectively for Trypanozoon [23] and Tbg [24]. Patients in whom trypanosomes were detected were treated according to the national procedures and WHO recommendations [25].

Epidemiological investigations

For each detected HAT case, a spatial and geo-referenced investigation of the travel history and living places was carried out to identify populations that share the same environment and are at increased risk for HAT [4, 26]. An epidemiological investigation was carried out to describe in detail the clinical evolution and the health and therapeutic itinerary of the patient (discussions with the family, discussions with health agents that examined the patient, consultation of all possible documentation such as the health book, the results of different analyses, hospitalisation certificates, prescriptions etc.)

Ethics

Before commencement, the DiTECT-HAT diagnostic trial (Diagnostic Tools for Human African Trypanosomiasis Elimination and Clinical Trials, work package 2, passive case detection) received ethical clearance (i) from the Comité Consultatif de Déontologie et d'Ethique of the French National Institute for research on sustainable development (IRD), (ii) from the Institutional Review Board of the Institute of Tropical Medicine Antwerp Belgium (reference 1133/ 16), (iii) from the Ethics Committee of the University of Antwerp (Belgian registration number B300201730927) and (iv) from the Comité National d'Ethique de la Recherche, Ministry of Public Health and Hygiene in Côte d'Ivoire (reference 076//MSHP/CNER-kp). The DiTECT-HAT trial has been registered at ClinicalTrials.gov, identifier NCT03356665. Before being included in DiTECT-HAT, the study was explained to the patient and her mother and written ascent and informed consent was obtained. A written ascent and informed consent was also obtained to study and describe the pathway of the HAT case in the frame of the HAT elimination project in Côte d'Ivoire for which ethical clearance was received from the Comité National d'Ethique de la Recherche, Ministry of Public Health and Hygiene in Côte d'Ivoire (reference 030-18/MSHP/CNER-kp).

Results

The patient was the first one diagnosed by the passive surveillance activities implemented in the two endemic foci in Côte d'Ivoire. The patient was an 11 years old girl detected in September 2017 in the Bayota CSU, a serological screening site in the Sinfra focus.

Diagnosis at inclusion

The girl presented with long-lasting fever that did not respond to malaria treatment, serious weight loss, weakness, sleep disruption, motor problems (tremor and unable to walk), anorexia and speaking difficulties. All three RDT performed in Bayota were positive (S1 Fig), and the girl was referred to the Sinfra HG for parasitological examination. In the absence of swollen cervical lymph nodes, parasitological confirmation was carried out by examination of blood in the mAECT, with only one trypanosome observed (S2 Fig). The patient was diagnosed with stage 2 HAT, having 160 white blood cells/µl and trypanosomes in the cerebrospinal fluid and was successfully treated using Nifurtimox Eflornithine Combination Therapy in the Bonon CSU since the Sinfra HG was, at that time, not yet functional for HAT treatment. Her dried blood spots were positive in immune trypanolysis, ELISA and LAMP but negative in real time PCR.

Geographical itinerary

The girl was born in January 2006 in Buyo, situated in South-West Cote d'Ivoire in the department of Soubré, in the Nawa region (Fig 2). Three weeks after her birth, her mother took her to Abidjan, from which 8 months later, they moved to Sinfra where they stayed for 5 years. In March 2011, she moved with the girl to Yabayo, a city in Central-West Côte d'Ivoire to do business. In September 2012, they went to Gagnoa, where the girl went to school and was under the supervision of her grandmother. The mother installed herself to work in Bouaké, a city in the centre of the country. In December 2015, the girl dropped out of school due to the deterioration of her health (see below). In January 2016, the mother left her job in Bouaké to seek health care for the girl in Abidjan where they stayed with the mother's family. Running out of financial means, they went to Bouaké in April 2017, for the mother to resume her job, while looking for a cure for the girl. After numerous examinations and despite a strong HAT suspicion established in Bouaké (details provided below), the mother, decided in August 2017 to take her girl to Bayota (20 km south of Sinfra) where her husband resided. It was at the Bayota CSU and then at the Sinfra HG that she was re-screened in September 2017 after which the diagnosis could be confirmed.

Clinical evolution and health seeking itinerary

The first clinical signs of HAT date back to October 2014 (Fig 2) with fever and headaches that were treated with traditional self-medication. In November 2014, the same symptoms persisted accompanied by general asthenia. The patient then consulted the nearest primary health care centre, the maternal and child protection centre (PMI) in Gagnoa where malaria was



Fig 2. Geographical and health itinerary of the patient between birth and HAT diagnosis. W = week(s), M = month(s), Y = year(s).

https://doi.org/10.1371/journal.pntd.0008588.g002

diagnosed and treated. Between December 2014 and July 2015, the PMI continued to diagnose and treat for malaria with monthly intervals, based on the same symptoms, which from June 2015 were accompanied by convulsions. This was the reason for visiting a traditional healer in August. Although from September to October 2015 all symptoms subsided, they reappeared in November 2015, together with hallucinations and behavioural problems (bulimia, hilarity, strong restlessness, abnormal movements). The patient was then hospitalized for 12 days in a private clinic in Gagnoa, but no improvement was observed. In December 2015, the patient's family called upon another traditional healer and a spiritual healer who attributed the troubles to a spirit called "DRAGON". Despite prayers and new treatments, symptoms worsened with appearance of limb paralysis.

In January 2016, the patient and her mother went to Abidjan and first consulted private structures. A strong anemia led to a sickle cell test, which turned out negative. Following the onset of severe fever and diarrhoea, typhoid fever was diagnosed and treated, and anti-anemic drugs were given. In April 2016, the existing symptoms persisted, with the addition of

abnormal movements, trembling of the limbs and sleep disturbances (daytime hypersomnolence and night time insomnia). The patient was then admitted to a private clinic. An analysis report and the patients' health record mention a HAT test, which consisted of a fresh blood examination, which was negative. Anti-anemics and antipyretics were again prescribed. In October 2016, the patient consulted a referral hospital of the tertiary level in Abidjan with, in addition to the aforementioned symptoms, eating difficulties. She was admitted to the neurology department and then to the trauma department. Leg massages were prescribed, as well as medicines for fever and poor appetite. Without improvement, in January 2017, the patient consulted another private clinic in Abidjan where she was diagnosed with malaria and treated.

Back in Bouaké in April 2017, the persistent symptoms associated with severe anemia made the patient to consult an urban health centre where she was once again prescribed antimalarials, antibiotics against typhoid fever and anti-anemics. In July 2017, with the worsening of motor disorders, a generalized paralysis (hypertonia of the limbs), anorexia, almost continuous hyper-somnolence and loss of consciousness, she was referred to the Bouaké CHU where she was admitted to the service of neurology. It is in this department that a precise clinical description was made, described elsewhere [27]. The first real clinical suspicion of HAT was made by a neurologist who knew the symptoms of the disease and who guided the patient to the Institut Pierre Richet, an institution of the National Institute of Public Health (INSP) specialized in research on vector-borne diseases including HAT. Although the RDT and the CATT were positive for presence of specific antibodies, the patient could not be confirmed by parasitological examination in the mAECT. Since a lumbar puncture was performed the day before at Bouaké CHU to perform a direct microscopic observation of cerebrospinal fluid (CSF) which turned out to be negative for trypanosomes, the lumbar puncture was not repeated at this occasion. The patient was invited to come back the next week to repeat the parasitological investigations. Unfortunately, the mother, who lost hope of recovery and trust in conventional medicine, decided to return to Bayota. It was thanks to the strong awareness-raising of the HAT team of IPR who staved in contact by phone with the mother and to the fact that Bayota was included in the HAT passive surveillance system that she finally agreed to go to the CSU of Bayota in September 2017. The confirmation of the serological suspicion of HAT convinced the mother to go to the Sinfra HG where her daughter was finally parasitologically confirmed.

Financial aspects

We made an estimation of the total expenditures made by the patient's family between the time of her first symptoms (October 2014) and her treatment (September 2017). We took into account the costs of consultations, the cost of laboratory analyses and the cost of drugs that were prescribed and purchased. We also estimated the mother's loss of income due to the interruption of her professional activities (Table 1). We did not include the transport costs and the costs of living outside the main residence. The total amount spent was 3 718 500 CFA francs or 5 669 Euros. For comparison, the gross domestic product per capita of Côte d'Ivoire was 1 383 Euros in 2017 (https://data.worldbank.org/indicator/NY.GDP.PCAP.CD?locations=CI).

Discussion

We examined the clinical, diagnostic, geographical, and financial aspects of a HAT case detected in Côte d'Ivoire. Only one single patient is described, which is clearly a limitation. Although interesting for its peculiarities, it is not possible to extrapolate some of the conclusions to the generality. However, the patient overall pathway clearly demonstrates the difficulties that can be encountered to arrive at the correct diagnosis within a HAT elimination context.

ITEM	Amount in FCFA*	Amount in EURO
Consultation costs	293 500	447,44
Drug costs	965 000	1 471,13
Costs of medical laboratory analyses	540 000	823,23
24 months' salary loss of the mother**	1 920 000	2 927,02
TOTAL	3 718 500	5 668,84

Table 1. Details of the expenses for case management of the patient between October 2014 and September 2017.

* FCFA = currency in Côte d'Ivoire (1 euro = 655,957 FCFA)

** The mother had to stop working for 24 months to take care of her daughter

https://doi.org/10.1371/journal.pntd.0008588.t001

Clinically, symptoms and signs for sleeping sickness are not specific [25]. The first symptoms observed (fever, headache, asthenia, anemia and digestive disorders) were attributed several times to malaria or typhoid fever, confirmed or not by laboratory analyses. The treatments prescribed during three years were in line with these diagnoses and mainly composed of antipyretics, antibiotics, antimalarials and anti-anemics, even after the onset of neurological disorders, since no alternative diagnosis could be made. Suspicion of HAT was raised for the first time in a private clinic in Abidjan, on the basis of sleep disruption, but as the diagnosis could not be confirmed at that time, suspicion was not maintained. It was only at a very advanced disease stage that an experienced neurologist strongly suspected HAT [27] and referred the patient to a specialized structure. The evolution of the patient's clinical signs with three years between the appearance of the first symptoms and an almost comatose state in which she was diagnosed, perfectly illustrates the chronic nature of HAT due to *Tbg* with a slow evolution and a progressive appearance of neurological disorders [28,29].

Also laboratory diagnosis [6], which is needed to initiate treatment for HAT, proved not to be easy. The first parasitological examination, the wet blood film examination, was negative but this technique has a poor sensitivity [30, 31]. The final confirmation was also retarded by 2 months due to a negative mAECT result, despite the fact that mAECT is the most sensitive parasitological method [32,33,34]. During the screening of the patient in September 2017, the three RDT were positive, but only one trypanosome could be detected in blood by the mAECT. This can be explained by the low parasite numbers present in the blood, which is characteristic for *Tbg* HAT, and shows the interest of repeating the parasitological examination [30]. The tests carried out on dried blood spots by the reference lab were positive with exception of the 18S and TgsGP PCR, which can be explained by the limited quantity of DNA on the filter paper [33,35]. In contrast to the blood results, many trypanosomes were observed upon cerebrospinal fluid examination, a phenomenon that is not uncommon in advanced stage HAT [33].

The travel history of the patient contributed to delay diagnosis as well. She spent her early childhood in Sinfra, a well-known HAT focus [3,36,37,38,39], where she was probably infected. Although Yabayo and Gagnoa are historic HAT foci, infection there seems less probable due to her short stay in Yabayo and an exclusively urban life as schoolgirl in Gagnoa, with little probability of exposure to tsetse flies. Unfortunately, the patient did not attend any active HAT screening session before leaving Sinfra, and her chances of being diagnosed, despite the onset of symptoms, were limited in Gagnoa where since 1994 no case of HAT had been detected [37].

From a service provider delay perspective, the patient consulted during 3 years at least twenty health centres in the public and private sector. She used the three levels of the national public health system pyramid and met with more than a hundred doctors, nurses, laboratory technicians and community health workers. This illustrates another difficulty, which comes from the fact that HAT is a neglected tropical disease, whose control, in the absence of vaccines or prophylactics, is based on case management often carried out by specialized teams on the edges of the national health system [40]. In Côte d'Ivoire, for example, control has long relied on two research institutions: Institut Pierre Richet in Bouaké and PRCT. Since 2006, PRCT has been the only centre for passive screening and treatment in the country [3,4,7]. This system offers only very limited access to specialized care, especially in rural areas where HAT mainly occurs [41]. It also limits the number of health workers capable of suspecting the disease, which in addition, is going down in a context of sharply decreasing prevalence. Today, HAT is very succinctly dealt with in the training of health personnel, whatever the level of training is. In addition, a large majority of health workers in Côte d'Ivoire considers HAT as a disease of the past century.

Unfortunately, this is also the case for at-risk communities who no longer consider the disease as a threat, and no longer participate in active screening sessions -hence their ineffectiveness [8] and who gradually lose the reflex of thinking about HAT. Neurological disorders are often attributed to a mystical cause and patients are referred to traditional and spiritual healers. Implication of traditional healers is probably common in diagnostic pathways of patients with rare diseases in an elimination context, but not as well documented for *Tbg* HAT [42], as for *rhodesiense* HAT [43], [44]. These steps contributed to patient delay in seeking the diagnosis. Actions should be taken to raise awareness of the traditional healers so that they can refer clinical suspects to health centres where HAT passive surveillance is implemented. Such an approach would compensate for the often-low presentation rate in health centres in remote areas that characterize the HAT foci [12].

The combination of all difficulties highlighted above, resulted in a patients and service provider delay which was much higher than the median time previously reported from Democratic Republic of Congo [45], which could possibly be attributed to the relatively high HAT prevalence at time of the study in DR Congo, compared to the elimination context in Côte d'Ivoire some years later. The delays led to late treatment of the patient who suffered a lot and still has neurological sequels when writing the manuscript. The present experience also illustrates the financial burden of HAT on families in particular when the suspicion index is low and diagnostic capacities are absent. The costs incurred, including the mothers salary loss for two years, were also much higher than the 44 US\$ previously described for DR Congo [45], and led the mother to seek for financial help from her family.

In a context of low HAT prevalence, the implementation of an effective passive surveillance integrated into the national health system [12] could help to overcome some of these difficulties. First, it reminds the health workers of the existence of HAT and gives them the means to screen for it using a simple protocol based on clinical and serological suspicion using RDT [46]. It makes care more accessible to all populations at risk, even the most remote, and targets clinical suspects even if they came to consult for other reasons. Finally, geographically localisation of confirmed patients, enables identification of the areas of transmission and improved orientation of the control efforts. Based on the clinical symptoms, the patient might have been diagnosed earlier, if a passive surveillance system had been in place. In Côte d'Ivoire, the national programme for elimination of HAT has decided to reinforce the passive surveillance system from 2018 on, by extending it to all hypo-endemic foci, which were defined as having one case detected between 2000 and 2015, and implemented it in 13 additional health centres. In addition to primary health care centres, passive surveillance has also been integrated in five tertiary level health centres including in the main psychiatric and neurological services (Abidjan, Bouaké and Bingerville). These centres receive patients from all over the country, as was the case for the patient who consulted two of them, and contribute to national surveillance.

This is important as the population at risk for HAT in West-Africa, is often characterized by significant mobility [7,36]. This population has in general very little chance to be tested for HAT when it leaves endemic areas.

Passive surveillance however has some limitations. It is based on clinical suspicion, and risks to detect patients mainly when they present with neurological symptoms and therefore are already in an advanced disease stage [7]. This is an important constraint, not only for the patient, but also for the community since the subject constitutes a reservoir of parasites and a source of contamination for a long time before being detected and treated. To detect additional cases as early as possible, passive surveillance should therefore be accompanied by adapted reactive active screening strategies, such as reactive door to door screening, [4], spatial tracking [26], follow-up of seropositive subjects [47,48,49,50] or identification of villages at risk [51]. It is also essential to inform the health personnel that, even if HAT prevalence has decreased, the disease still exists and carries a risk to re-emerge [52]. Health personnel needs to be trained continuously and must receive the means to diagnose the disease. Equally important, public awareness on HAT should be raised, at the local level in the areas at risk but also at national level, including information on the possibility to be screened in primary health centres. Difficulties in passive case detection of HAT in the context of a significant drop in prevalence have also been observed in DR Congo [53]. In this country, the most affected by HAT [2], a national HAT commemoration day has been created [54], a sensitization initiative that could be extended to all endemic countries.

In conclusion, the detailed analysis of the clinical, diagnostic, geographical, and financial aspects of this HAT case, whose diagnosis was facilitated by a clinical trial, underlines both the complexity and the importance of detecting the last HAT cases and of reaching HAT elimination in Côte d'Ivoire where tsetse flies are found throughout the territory [55]. Reinforcement of passive surveillance at national scale, integrated in the existing health system could help to overcome some difficulties in combination with continued sensitization and health staff training.

Supporting information

S1 Fig. Photo of the three positive rapid diagnostic tests performed at the time of HAT diagnosis in September 2017. (TIF)

S2 Fig. Video of the single trypanosome detected by the mAECT parasitological test confirming the HAT status of the patient in September 2017. (MP4)

Acknowledgments

We acknowledge all the technicians from the HAT teams of Institut Pierre Richet of Bouaké, University Jean Lorougnon Guédé of Daloa, Projet de Recherches Cliniques sur la Trypanosomiase of Daloa, Centre de Santé Urbain of Bonon, Hôpital Général de Sinfra and HAT National Elimination Program of Côte d'Ivoire. We also acknowledge all the health agents involved in the HAT passive surveillance in Côte d'Ivoire.

Author Contributions

Conceptualization: Bruno Bucheton, Philippe Büscher, Veerle Lejon, Vincent Jamonneau.

Funding acquisition: Veerle Lejon.

- **Investigation:** Minayégninrin Koné, Emmanuel Kouassi N'Gouan, Dramane Kaba, Louis N'Dri, Cyrille Mambo Kouamé, Valentin Kouassi Nanan, Gossé Apollinaire Tapé, Bamoro Coulibaly, Fabrice Courtin, Bernardin Ahouty, Vincent Djohan.
- Methodology: Minayégninrin Koné, Emmanuel Kouassi N'Gouan, Mathurin Koffi, Lingué Kouakou, Louis N'Dri, Cyrille Mambo Kouamé, Valentin Kouassi Nanan, Gossé Apollinaire Tapé, Bernardin Ahouty, Vincent Djohan.
- Project administration: Dramane Kaba, Philippe Solano, Veerle Lejon, Vincent Jamonneau.
- Supervision: Dramane Kaba, Mathurin Koffi, Lingué Kouakou, Philippe Solano, Philippe Büscher, Veerle Lejon, Vincent Jamonneau.
- Writing original draft: Minayégninrin Koné, Dramane Kaba, Mathurin Koffi, Bamoro Coulibaly, Fabrice Courtin, Veerle Lejon.
- Writing review & editing: Bruno Bucheton, Philippe Büscher, Vincent Jamonneau.

References

- Büscher P, Cecchi G, Jamonneau V, Priotto G. Human African trypanosomiasis. Lancet. 2017; 390: 2397–2409. https://doi.org/10.1016/S0140-6736(17)31510-6 PMID: 28673422
- Franco JR, Cecchi G, Priotto G, Paone M, Diarra A, Grout L, et al. Monitoring the elimination of human African trypanosomiasis: Update to 2016. PLoS Negl Trop Dis. 2018; 12: e0006890. https://doi.org/10. 1371/journal.pntd.0006890 PMID: 30521525
- Kaba D, Djè NN, Courtin F, Oke E, Koffi M, Garcia A, et al. L'impact de la guerre sur l'évolution de la THA dans le centre-ouest de la cote d'ivoire. Trop Med Int Health. 2006; 11: 136–143. <u>https://doi.org/ 10.1111/j.1365-3156.2005.01549.x PMID: 16451337</u>
- Koffi M, N'Djetchi M, Ilboudo H, Kaba D, Coulibaly B, N'Gouan E, et al. A targeted door-to-door strategy for sleeping sickness detection in low-prevalence settings in Côte d'Ivoire. Parasite. 2016; 23: 51. https://doi.org/10.1051/parasite/2016059 PMID: 27849517
- Magnus E, Vervoort T, Van Meirvenne N. A card-agglutination test with stained trypanosomes (CATT) for the serological diagnosis of T. b. gambiense trypanosomiasis. Ann Soc Bel Med Trop. 1978; 58: 169–176.
- Chappuis F, Loutan L, Simarro P, Lejon V, Büscher P. Options for field diagnosis of human African trypanosomiasis. Clin Microbiol Rev. 2005; 18: 133–146. https://doi.org/10.1128/CMR.18.1.133-146.2005 PMID: 15653823
- Kambiré R, Lingué K, Courtin F, Sidibé I, Kiendrébéogo D, N'gouan KE, et al. La Trypanosomiase Humaine Africaine dans l'espace ivoiro-burkinabé: optimisation des stratégies de surveillance épidémiologique. Parasite. 2012; 19: 389–396. https://doi.org/10.1051/parasite/2012194389 PMID: 23193524
- 8. Robays J, Bilengue MMC, Stuyft PV der, Boelaert M. The effectiveness of active population screening and treatment for sleeping sickness control in the Democratic Republic of Congo. Trop Med Int Health. 2004; 9: 542–550. https://doi.org/10.1111/j.1365-3156.2004.01240.x PMID: 15117297
- Robays J, Lefèvre P, Lutumba P, Lubanza S, Kande Betu Ku Mesu V, Van der Stuyft P, et al. Drug toxicity and cost as barriers to community participation in HAT control in the Democratic Republic of Congo. Trop Med Int Health. 2007; 12: 290–298. https://doi.org/10.1111/j.1365-3156.2006.01768.x PMID: 17300638
- Mitashi P, Hasker E, Lejon V, Kande V, Muyembe J-J, Lutumba P, et al. Human African trypanosomiasis diagnosis in first-line health services of endemic countries, a systematic review. PLoS Negl Trop Dis. 2012; 6: e1919. https://doi.org/10.1371/journal.pntd.0001919 PMID: 23209860
- Franco JR, Simarro PP, Diarra A, Jannin JG. Epidemiology of human African trypanosomiasis. Clin Epidemiol. 2014; 6: 257–275. https://doi.org/10.2147/CLEP.S39728 PMID: 25125985
- Mitashi P, Hasker E, Mbo F, Geertruyden JPV, Kaswa M, Lumbala C, et al. Integration of diagnosis and treatment of sleeping sickness in primary healthcare facilities in the Democratic Republic of the Congo. Trop Med Int Health. 2015; 20: 98–105. https://doi.org/10.1111/tmi.12404 PMID: 25329353
- 13. Kroa E, Diaby R, Niaré A, Traoré Y, Ahoussou EM, Yao GSA, et al. Analyse de la collaboration entre médecines traditionnelle et moderne dans la région du Sud Bandama (Côte d'Ivoire). Revue du Cames-Série Pharm Méd Trad Afr. 2014; 17: 21–27.

- Lumbala C. C, Bisser S, Nguertoum, Flevaud L., Jacquet D. Development and evaluation of a rapid screening test for sleeping sickness. Ann Afr Méd. 2013; 6, Suppl. 1: 49.
- Büscher P, Mertens P, Leclipteux T, Gilleman Q, Jacquet D, Mumba-Ngoyi D, et al. Sensitivity and specificity of HAT Sero-K-SeT, a rapid diagnostic test for serodiagnosis of sleeping sickness caused by *Trypanosoma brucei gambiense*: a case-control study. Lancet Glob Health. 2014; 2: e359–e363. https://doi.org/10.1016/S2214-109X(14)70203-7 PMID: 25103304
- Büscher P, Gilleman Q, Lejon V. Rapid diagnostic test for sleeping sickness. N Engl J Med. 2013; 368: 1069–1070. https://doi.org/10.1056/NEJMc1210373 PMID: 23484849
- Lumsden WHR, Kimber CD, Evans DA, Doig SJ. *Trypanosoma brucei*: miniature anion-exchange centrifugation technique for detection of low parasitaemias: adaptation for field use. Trans R Soc Trop Med Hyg. 1979; 73: 312–317. https://doi.org/10.1016/0035-9203(79)90092-0 PMID: 473329
- Büscher P, Ngoyi DM, Kaboré J, Lejon V, Robays J, Jamonneau V, et al. Improved models of mini anion exchange centrifugation technique (mAECT) and modified single centrifugation (MSC) for sleeping sickness diagnosis and staging. PLoS Negl Trop Dis. 2009; 3: e471. <u>https://doi.org/10.1371/journal.pntd.0000471</u> PMID: 19936296
- Van Meirvenne N, Magnus E, Büscher P. Evaluation of variant specific trypanolysis tests for serodiagnosis of human infections with *Trypanosoma brucei gambiense*. Acta Trop. 1995; 60: 189–199. https:// doi.org/10.1016/0001-706x(95)00127-z PMID: 8907397
- Lejon V, Büscher P, Magnus E, Moons A, Wouters I, Van Meirvenne N. A semi-quantitative ELISA for detection of *Trypanosoma brucei gambiense* specific antibodies in serum and cerebrospinal fluid of sleeping sickness patients. Acta Trop. 1998; 69: 151–164. https://doi.org/10.1016/s0001-706x(97) 00137-x PMID: 9588235
- Hasker E, Lutumba P, Mumba D, Lejon V, Büscher P, Kande V, et al. Diagnostic accuracy and feasibility of serological tests on filter paper samples for outbreak detection of *T.b. gambiense* human African trypanosomiasis. Am J Trop Med Hyg. 2010; 83: 374–379. https://doi.org/10.4269/ajtmh.2010.09-0735 PMID: 20682885
- Njiru ZK. Rapid and sensitive detection of human African trypanosomiasis by loop-mediated isothermal amplification combined with a lateral-flow dipstick. Diagn Microbiol Infect Dis. 2011; 69: 205–209. https://doi.org/10.1016/j.diagmicrobio.2010.08.026 PMID: 21251567
- Deborggraeve S, Claes F, Laurent T, Mertens P, Leclipteux T, Dujardin JC, et al. Molecular dipstick test for diagnosis of sleeping sickness. J Clin Microbiol. 2006; 44: 2884–2889. <u>https://doi.org/10.1128/JCM.</u> 02594-05 PMID: 16891507
- Radwanska M, Claes F, Magez S, Magnus E, Perez-Morga D, Pays E, et al. Novel primer sequences for polymerase chain reaction-based detection of *Trypanosoma brucei gambiense*. Am J Trop Med Hyg. 2002; 67: 289–295. https://doi.org/10.4269/ajtmh.2002.67.289 PMID: 12408669
- WHO. Control and Surveillance of Human African Trypanosomiasis: Report of a WHO Expert Committee. World Health Organization; 2013.
- Courtin F, Jamonneau V, Camara M, Camara O, Coulibaly B, Diarra A, et al. A geographical approach to identify sleeping sickness risk factors in a mangrove ecosystem. Trop Med Int Health. 2010; 15: 881– 889. https://doi.org/10.1111/j.1365-3156.2010.02559.x PMID: 20545924
- Bony KE, Akani AF, Kaba D, Gnazegbo A, Diakite I, Karidioula HA, et al. Mouvements anormaux du sujet jeune et trypanosomiase humaine africaine: un couple presque oublié. Prat Neurol-FMC. 2019; 10: 162–166.
- Blum J, Schimid C, Burri C. Clinical aspects of 2541 patients with second stage human African trypanosomiasis. Acta trop. 2006; 97: 55–64. <u>https://doi.org/10.1016/j.actatropica.2005.08.001</u> PMID: 16157286
- Checchi F, Filipe JAN, Barrett MP, Chandramohan D. The natural progression of gambiense sleeping sickness: what is the evidence? PLoS Negl Trop Dis. 2008; 2: e303 https://doi.org/10.1371/journal. pntd.0000303 PMID: 19104656
- Miezan T, Meda A, Doua F, Cattand P. Evaluation of the parasitologic technics used in the diagnosis of human Trypanosoma gambiense trypanosomiasis in the Ivory Coast. Bull Soc Pathol Exot. 1994; 87: 101–104. PMID: 8061525
- Lutumba P, Robays J, Miaka C, Kande V, Mumba D, Büscher P, et al. Validity, cost and feasibility of the mAECT and CTC confirmation tests after diagnosis of African of sleeping sickness. Trop Med Int Health. 2006; 11: 470–478. https://doi.org/10.1111/j.1365-3156.2006.01591.x PMID: 16553930
- Camara M, Camara O, Ilboudo H, Sakande H, Kaboré J, N'Dri L, et al. Sleeping sickness diagnosis: use of buffy coats improves the sensitivity of the mini anion exchange centrifugation test. Trop Med Int Health. 2010; 15: 796–799. https://doi.org/10.1111/j.1365-3156.2010.02546.x PMID: 20497407

- **33.** Mumba D, Ekangu R, Mumvemba K, Pyana P, Balharbi F, Decq M, et al. Performance of parasitological and molecular techniques for the diagnosis and surveillance of gambiense sleeping sickness. PLoS Negl Trop Dis. 2014; 8: e2954. https://doi.org/10.1371/journal.pntd.0002954 PMID: 24921941
- 34. Lejon V, Büscher P, Nzoumbou-Boko R, Bossard G, Jamonneau V, Bucheton B, et al. The separation of trypanosomes from blood by anion exchange chromatography: From Sheila Lanham's discovery 50 years ago to a gold standard for sleeping sickness diagnosis. PLoS Negl Trop Dis. 2019; 13: e7051.
- Deborggraeve S, Büscher P. Recent progress in molecular diagnosis of sleeping sickness. Expert Rev Mol Diagn. 2012; 12: 719–730. https://doi.org/10.1586/erm.12.72 PMID: 23153239
- Laveissière C, Sané B, Diallo PB, Truc P, Méda AH. Le risque épidémiologique dans un foyer de la maladie du sommeil en Côte d'Ivoire. Trop Med Int Health. 1997; 2: 729–732. https://doi.org/10.1046/j. 1365-3156.1997.d01-378.x PMID: 9324419
- Dje NN, Miezan TW, N'guessan P, Brika P, Doua F, Boa F. Distribution géographique des trypanosomés pris en charge en Côte d'Ivoire de 1993 à 2000. Bull Soc Pathol Exot. 2002; 95: 359–361. PMID: 12696376
- Jamonneau V, Garcia A, Ravel S, Cuny G, Oury B, Solano P, et al. Genetic characterization of *Trypanosoma brucei gambiense* and clinical evolution of human African trypanosomiasis in Cote d'Ivoire. Trop Med Int Health. 2002; 7: 610–621. https://doi.org/10.1046/j.1365-3156.2002.00905.x PMID: 12100445
- Berté D, De Meeûs T, Kaba D, Séré M, Djohan V, Courtin F, et al. Population genetics of *Glossina pal-palis palpalis* in sleeping sickness foci of Côte d'Ivoire before and after vector control. Inf Genet Evol. 2019; 75: 103963.
- Laveissière C, Sané B, Méda HA. Measurement of risk in endemic areas of human African trypanosomiasis in Côte d'Ivoire. Trans R Soc Trop Med Hyg. 1994; 88: 645–648. <u>https://doi.org/10.1016/0035-</u> 9203(94)90207-0 PMID: 7886756
- Simarro PP, Cecchi G, Franco JR, Paone M, Diarra A, Ruiz-Postigo JA, et al. Mapping the capacities of fixed health facilities to cover people at risk of gambiense human African trypanosomiasis. Int J Health Geogr. 2014; 13: 4. https://doi.org/10.1186/1476-072X-13-4 PMID: 24517513
- **42.** Dama E, Drabo A, Kaboré J, Ouédraogo E, Coulibaly B, Ilboudo H, et al. Description of the first native sleeping sickness case diagnosed in Burkina Faso in the over two decades. PLoS Negl Trop Dis. 2018; 12: e6677.
- 43. Odiit M, Shaw A, Welburn SC, Fèvre EM, Coleman PG, Mc Dermot JJ. Assessing the patterns of health seeking behaviour and awareness among sleeping sickness patients in Eastern Uganda. Ann Trop Med Parasitol. 2004; 98: 339 348. https://doi.org/10.1179/000349804225003389 PMID: 15228715
- 44. Bukachi SA, Wandibba S, Nyamongo IK. The treatment pathways followed by cases of human African trypanosomiasis in western Kenya and eastern Uganda. Ann Trop Med Parasitol. 2009; 103: 211–220. https://doi.org/10.1179/136485909X398230 PMID: 19341536
- 45. Hasker E, Lumbala C, Mbo F, Mpanya A, Kande V, Lutumba P, et al. Health care-seeking behaviour and diagnostic delays for human African trypanosomiasis in the Democratic Republic of the Congo. Trop Med Int Health. 2011; 16: 869–874. <u>https://doi.org/10.1111/j.1365-3156.2011.02772.x</u> PMID: 21447063
- 46. Jamonneau V, Bucheton B. The challenge of serodiagnosis of sleeping sickness in the context of elimination. Lancet Glob Health. 2014; 2: e306–e307. <u>https://doi.org/10.1016/S2214-109X(14)70226-8</u> PMID: 25103289
- 47. Garcia A, Jamonneau V, Magnus E, Laveissière C, Lejon V, N'guessan P, et al. Follow-up of Card Agglutination Trypanosomiasis Test (CATT) positive but apparently aparasitaemic individuals in Côte d'Ivoire: evidence for a complex and heterogeneous population. Trop Med Int Health. 2000; 5: 786–793. https://doi.org/10.1046/j.1365-3156.2000.00623.x PMID: 11123826
- Ilboudo H, Jamonneau V, Camara M, Camara O, Dama E, Léno M, et al. Diversity of response to Trypanosoma brucei gambiense infections in the Forecariah mangrove focus (Guinea): perspectives for a better control of sleeping sickness. Microbes Infect. 2011; 13: 943–952. https://doi.org/10.1016/j.micinf. 2011.05.007 PMID: 21658462
- 49. Ilboudo H, Jamonneau V, Koffi M, Kaboré J, Amoussa R, Holzmuller P, et al. Trypanosome-induced Interferon-γ production in whole blood stimulation assays is associated with latent *Trypanosoma brucei* gambiense infections. Microbes Infect. 2016; 18: 436–440. https://doi.org/10.1016/j.micinf.2016.03.002 PMID: 26993030
- Ahouty B, Koffi M, Hamidou I, Simo G, Matovu A, Mulindwa J, et al. Candidate genes-based investigation of susceptibility to Human African Trypanosomiasis in Côte d'Ivoire. PLoS Negl Trop Dis. 2017; 11: e5992.
- Courtin F, Camara O, Camara M, Kagbadouno M, Bucheton B, Solano P, et al. Sleeping sickness in the historical focus of forested Guinea: update using a geographically based method. Parasite. 2019; 26: 61. https://doi.org/10.1051/parasite/2019061 PMID: 31599229

- 52. Camara M, Ouattara E, Duvignaud A, Migliani R, Camara O, Leno M, et al. Impact of the Ebola outbreak on Trypanosoma brucei gambiense infection medical activities in coastal Guinea, 2014–2015: A retrospective analysis from the Guinean national Human African Trypanosomiasis control program. PLoS Negl Trop Dis. 2017; 11: e6060.
- **53.** Mudji J, Benhamou J, Mwamba-Miaka E, Burri C, Blum J. The Flipside of eradicating a disease; Human African Trypanosomiasis in a Woman in Rural Democratic Republic of Congo: A Case Report. Trop Med Infect Dis. 2019; 4: 142.
- 54. DNDi: Drugs for Neglected Diseases initiative, The Celebration of the first anniversary National HAT Day in DRC. 2019 http://www.forumdesas.org/spip.php?article18469 -
- 55. Nekpeni EB, Dagnogo M, Eouzan JP. Détermination de la limite géographique entre deux sousespèces de glossines en Côte d'Ivoire: Glossina palpalis palpalis (Robineau-Desvoidy, 1830) et G.P. gambiensis (Vanderplank, 1949). Trop. Med. Parasit. 1988; 40: 12–15.